Author’s response to reviews

Title: Determination of the optimal inspiratory pressure providing adequate ventilation while minimizing gastric insufflation using real-time ultrasonography in Chinese children: a prospective, randomized, double-blind study

Authors:

Xiaowei Qian (qxwzhejiagedu@163.com)
Qiong Hu (huqiong_1221@qq.com)
Hang Zhao (zhaohang17777@163.com)
Bo Meng (mengbo1206@163.com)
Yang Nan (nay0000@163.com)
Hong Cao (Caohongwz@163.com)
Qingquan Lian (lianqingquanmz@163.com)
Jun Li (lijunwzmu@126.com)

Version: 4 Date: 17 Jul 2017

Author’s response to reviews:

Dear Prof. Spadaro

We appreciate editor and reviewers very much for their constructive comments and suggestions on our manuscript entitled “Determination of the optimal inspiratory pressure providing adequate ventilation while minimizing gastric insufflation using real-time ultrasonography in Chinese children: a prospective, randomized, double-blind study”. (ID: BANE-D-17-00041R2).

We have studied reviewer’s comments carefully and have made revision which marked in red in the paper. Attached please find the revised version, which we would like to submit for your kind consideration. I am sorry that we forgot to add the funds which support this work. We added the funds this time. In addition, Dr. Qian wants to change the order of his work institution. We would like to express our great appreciation to you and reviewers for comments on our paper. Looking forward to hearing from you.
Thank you and best regards.

Yours sincerely,

Jun Li

List of Responses

(Reviewer 1): Thank you for your revision. Unfortunately I am not able to read the modified flowchart due to its low quality (when I try to read it by enlarging it, it is impossible for me to appreciate written words). Please submit a figure with a better quality.

Answer: We have modified the flowchart. Because there are five groups and the PDF transfer might decrease the quality, I try my best to make a high quality. Thanks!